Author's response to reviews

Title: DR_SEQAN: a PC/Windows-based software to evaluate drug resistance using human immunodeficiency virus type 1 genotypes

Authors:

Cesar Garriga (cgarriga@msc.es)
Luis Menendez-Arias (lmenendez@cbm.uam.es)

Version: 2 Date: 16 August 2005

Author's response to reviews:

Dear Editor,

In response to your e-mail of August 9th, suggesting that this manuscript would be better suited to BMC Infectious Diseases, we have agreed to transfer it to that journal.

According to instructions to authors for software articles in BMC Infectious Diseases, we have included the 'Competing interests' section in page 9 of the revised manuscript.

We have not made any additional changes in the Abstract and Manuscript sections, since software papers have the same abstract and manuscript organization in both journals (BMC Bioinformatics and BMC Infectious Diseases).

We believe that the required changes have been made, but if you want us to make any additional modification, please let us know.

Best regards,

The authors